Author's response to reviews

Title: Changes in bone marrow lesions in response to weight loss in obese knee osteoarthritis patients: A prospective cohort study

Authors:

Henrik Gudbergsen (drhg@dadlnet.dk)
Mikael Boesen (mikael.boesen@gmail.com)
Robin Christensen (robin.christensen@frh.regionh.dk)
Else M Bartels (emb@frh.regionh.dk)
Marius Henriksen (marius.henriksen@frh.regionh.dk)
Bente Danneskiold-Samsøe (bds@frh.regionh.dk)
Henning Bliddal (henning.bliddal@frh.regionh.dk)

Version: 4 Date: 23 February 2013

Author's response to reviews: see over
Dear Editor.
Thank you for the input given by yourself and the reviewers.

With respect to the reviewers comments I have addressed these in the following way referring to line numbers in the manuscript document containing track changes (manuscript with track changes.21feb2013).
The manuscript document containing track changes has been uploaded under the section named “Additional material files” on the webpage. The manuscript document without track changes has been uploaded under the section named “Main manuscript” on the webpage.

Reviewer 1

Reviewer comment: Adding text referring to the sensitivity analysis.
Author response: Agree.
Author action: Please see lines 292-293 and 488-489.